Author’s response to reviews

Title: Ethical aspects of diagnosis and interventions for children with Fetal Alcohol Spectrum Disorder (FASD) and their families

Authors:

Gert Helgesson (gert.helgesson@ki.se)
Göran Bertilsson (Goran.Bertilsson@sbu.se)
Helena Domeij (helena.domeij@sbu.se)
Gunilla Fahlström (gunilla.fahlstrom@sbu.se)
Emelie Heintz (emelie.heintz@sbu.se)
Anders Hjern (anders.hjern@ki.se)
Christina Nehlin Gordh (Christina.Nehlin.Gordh@neuro.uu.se)
Viviann Nordin (viviann.nordin@ki.se)
Jenny Rangmar (jenny.rangmar@psy.gu.se)
Ann-Margret Rydell (annmargret.rydell@psyk.uu.se)
Viveka Sundelin-Wahlsten (Viveka.Sundelin@neuro.uu.se)
Monica Hultcrantz (Monica.Hultcrantz@sbu.se)

Version: 2 Date: 25 Aug 2017

Author’s response to reviews:

Thanks for giving us this opportunity to come back with a re-resubmission. We are particularly grateful that you insisted to have the language checked since the overall improvement is considerable thanks to a series of minor changes.

This is how we respond to the additional requirements:

1. Lack of formal inclusion criteria for the manuscripts to be included should be mentioned as a limitation.

RESPONSE: A couple of sentences pointing this out have been added at the very beginning of Limitations (see first few lines).
2. The final sentence of the abstract does not make sense.

RESPONSE: Our point was poorly stated. This sentence has now been rewritten.

3. The conclusion now lacks context. ... You also need a final concluding sentence.

RESPONSE: The conclusion is rewritten (affecting all paragraphs in the section) in order to meet the criticism.

4. Please have your paper proof-read for English expressions.

RESPONSE: The instances identified in your reply have been corrected. In addition, we have had the text read through and language edited by a native English-speaking colleague, resulting in minor changes spread throughout the text.

Regarding points 3 and 4: If needed, we can match the submitted first revised version with the present one, thus identifying all changes between the documents in a track-changes fashion.